Author’s response to reviews

Title: Three Years of Pulmonary Rehabilitation: Inhibit the Decline in Airflow Obstruction, Improve Exercise endurance Time, and Body-Mass Index, in Chronic Obstructive Pulmonary Disease

Authors:

David Stav (dstav@asaf.health.gov.il)
Meir Raz (raz_m@maccabi.org.il)

Version: 2 Date: 1 November 2008

Author’s response to reviews: see over
Dear Dr. Norton  
Chief Editor  
BMC Pulmonary Medicine

The enclosed paper shows an additional important advantage of pulmonary rehabilitation, which was not fully disclosed. The main reason because rehabilitation programs are usually shorter. Since Dr. Gosselink who is on the editorial board disagree with this concept, and is negatively biased, I wish he wouldn't be in the decision team.  
Sincerely Yours

David Stav MD